



#### ESS weekly meeting (2022 W05)

A. Miyazaki et al





- Optimistic planning without any contingency
- We aim at testing three modules before the Easter holidays



#### W04 & W05 progress & W06 planning



wee	k W04												
date		MON		TUE		WED		THU		FRI		SAT	SUN
		24-jan		25-jan		26-jan		27-jan		28-jan		29-jan	30-jan
		m	а	m	а	m	а	m	а	m	а		
present CM	СМ07	4K filling; 2K pumping	CTS test	piezo ano	d heat load	CTS test with hea	n Beckhoff PLC; ht load	heat load			start warminន្	vent insulation vacuum	warming up
next CM	СМ08												
next next CM	СМ09	preparation at Orsay											

week		W05											
date		MON		TUE		WED		THU		FRI		SAT	SUN
		31-jan		01-feb		02-feb		03-feb		04-feb		05-feb	06-feb
		m	а	m	а	m	а	m	а	m	а		
previous CM	СМ07	warming up completed	disconnect vac pumps	disconnect cryogenics	swap modules	N2 filling	out going test		waiting in the b	ж	activate shock sensor	waiting	in the box
present CM	CM08	doorknob mounting				waveguide	e connection	connect cryo connect lines vacuum pumps		pumping vacuum			
next CM	СМ09				preparat	ion at Orsay		departure from Orsay transport				isport	

#### We are here

wee	ek 🛛	W06											
date		MON		TUE		WED		THU		FRI		SAT	SUN
		07-feb		08-feb		09-feb		10-feb		11-f	eb	12-feb	13-feb
		m	а	m	а	m	а	m	а	m	а		
previous CM	СМ07	departur	re to ESS	preparing report				publish report					
present CM	СМ08			coupler warm conditioning						purging	purging N2 cooling		
next CM	СМ09	transport from Orsay						arriv	al at UU	reception test			

#### What time?



## CM07: LT02 malfunctioning confirmed





- LT02 is touching the ground only at cold
- At warm, no problem was observed
- Similar observation in the elliptical modules (IFJPAN)
- Is it possible to fix this in ESS?



## CM07: EH02 and EH20 seem swapped inside



- EH02 heated up TT20 indicating EH20 is wrongly connected
- No issues were found outside the module (CM08 was OK)
- Cable swapping inside
  CM07 similar to the
  first CM02 test

NOW

19:00

< >

AUTO

22:00



#### CM07: CAVIN reached 12 MV/m



Mild field emission was observed





FREIA



Ch0

Ch1

Ch0

🗹 Ch1

Ch2

Ch3

352

Output enabled

Output delay

Output delay delta

0 0.5 1 1.5 2 2.5 3 3.5 4 4.5 5 5.5 6

0 0,5 1 1,5 2 2,5 3 3,5 4 4,5 5 5,5 6

0

- 0 ns

<u></u>0 ps Measure Tau at Time

32200

Tau [µs]

160.941

Tau set

Quench\_Warning

0

120

Enable

Reset Quench Warning

Eacc\_pk\_Pt

PT10

1E-9

12,0634

Eacc\_pk\_Pf

8 -

15,9547

PT20

2,6E-9

Pf\_max (W) Q0 Dynamic Eacc Dynamic

PT02

9999

0

PT03

31,4

124

0

TT07

2,05486

346765

TT05

2,01885

#### CM07: CAVOUT reached 12 MV/m





#### Systematic issue in static heat load





Dynamic heat load is a relative measurement so reported small values are not affected by this findings in static heat load



#### CM07: out-gassing during warming up



time [hour]

FREI









# Cf. Out gassing during coupler conditioning





### CM07: almost ready for shipping







# CM08 preparation for coupler conditioning



- The flange for CM02 did not fit so we open the CM08 pallet and fit the flange
- The fitting is very tight. Something changed before CM06 and after CM07?